Author’s response to reviews

Title: Quantitative proteomic characterization of human sperm cryopreservation: using dataindependent acquisition (DIA) mass spectrometry

Authors:

Longlong Fu (fullpcm@163.com)
Qi An (276724023@qq.com)
Kaishu Zhang (1323961050@qq.com)
Ying Liu (1151294240@qq.com)
Yue Tong (fullcmu@126.com)
Jianfeng Xu (jianfengxu2016@163.com)
Fang Zhou (zhoufang16@126.com)
Xiaowei Wang (xwwang16@163.com)
Yin Guo (495424764@qq.com)
Wenhong Lu (wenhonglu16@163.com)
Xiaowei Liang (liangxw61@126.com)
Yiqun Gu (guyq16@163.com)

Version: 6 Date: 25 Nov 2019

Author’s response to reviews:

Many thanks for your comments.

1. Authors' Contributions

Answer: WHL and XWL verify and interpretation all the data. And them and YQG revised the manuscript.

2. Change of authorship form

Answer: We uploaded the new form again.